Dear Editor

Thank you considering our manuscript "Potential risk factors of diabetic neuropathy: a cross sectional study" (MS#1315091906736316) for publication in your journal. As you requested English language of the manuscript was improved using a copyediting service (Manuscript Presentation Service). The manuscript was checked again and modified considering manuscript formatting checklist. Also minor revisions are as follow:

Reviewer#2: Satoshi Kuwabara

-P value of 0.00 means the same p<0.001 and was edited in the text and table.

Reviewer#3: Peter J Dyck

-Although mean HbA1C was not significantly different between cases and controls but after classification of diabetes control based on average HbA1C levels, multivariate logistic regression analysis revealed significant association between diabetes control (according average HbA1C levels) and diabetic neuropathy as shown in table 3.

Best Regards,

Bagher Larijani